**Author’s response to reviews**

**Title:** Disease burden and related medical costs of rotavirus infections in Taiwan

**Authors:**

Chun-Yi Lu (chunyi@ha.mc.ntu.edu.tw)
Tsai-Ling Lauderdale (lauderdale@nhri.org.tw)
Yin-Hua Fang (yinna@ha.mc.ntu.edu.tw)
Chung-Yi Wang (tucle91@hotmail.com)
Yu-Huai Ho (hoyuhuai@yahoo.com.tw)
Che-Lun Hung (ly7077@tpts6.seed.net.tw)
Luan-Yin Chang (allen0104.tw@yahoo.com.tw)
Chin-Yun Lee (mfsun@ha.mc.ntu.edu.tw)
Li-Min Huang (lmhuang@ha.mc.ntu.edu.tw)

**Version:** 11  **Date:** 21 November 2006

**Author’s response to reviews:**

Dear Editors of the BMC Infectious Diseases:

Enclosed please find the final version of our manuscript entitled "Disease burden and related medical costs of rotavirus infections in Taiwan" (MS: 1882125533102107). The manuscript has been modified based on the formatting changes requested in the pre-accept e-mail we received on November 14. Below is a summary of the formatting changes:

1. Running title has been deleted
2. All websites have been removed from the text and put into the References section and the reference numbers have been changed accordingly for the manuscript.
3. All authors have read and approved the final manuscript. This statement has been added to the Author's contribution section. This study was not funded by any grants and was carried out as part of our routine work, thus no additional information was added to the acknowledgements section.

The following changes were also made on the manuscript. The first two sentences under Data from the BNHI database in Methods section were removed since the same sentences are in Introduction already. Normalized economic burden indexes in Table 5 were recalculated to make the data more precise with explanations added to the footnote. This did not affect our study or conclusion. Minor English revisions were made to improve the readability.

We trust that these modifications meet your journal's requirement and look forward to seeing our manuscript published in BMC Infectious Diseases. Thank you.

Sincerely yours,

Tsai-Ling Lauderdale
Division of Clinical Research
National Health Research Institutes
35 Keyan Road. Zhunan Town
Miaoli County, Taiwan 350